Author's response to reviews

Title: Toxoplasmosis in the immunocompetent patient: a case report

Authors:

    Aneta K Taila (akouza1@hfhs.org)
    Ameet S Hingwe (ahingwe@yahoo.com)
    Laura E Johnson (ljohns14@hfhs.org)

Version: 2 Date: 8 December 2009

Author's response to reviews: see over
December 8, 2009

Dear Editor:

Attached please find our manuscript entitled “Toxoplasmosis in the Immunocompetent Patient: A Case Report” to be considered for the Case reports section of the Journal of Medical Case reports.

We had initially submitted the case to Cases Journal – but unfortunately after 6 weeks were told that the journal was no longer publishing. This case illustrates a challenging diagnosis and points towards the fact that history taking is important in any patient encounter.

Cases Journal had kindly agreed for a concession fee for publication if the article would have been accepted and we sincerely request the same for JMCR.

We feel the information contained in the manuscript will be of interest to readers of the above journal.

We thank you for your consideration of this manuscript.

Sincerely,

Ameet S Hingwe
Aneta K Taila
Laura E Johnson
Division of Infectious Diseases, Department of Pharmacy
Henry Ford Health System.